Doctoral Thesis (Abridged)

博士論文 (要約)

The Health-Economic Burden of Climate Change and Climate Variability: An Empirical Study from Uganda

(気候変動・気象変化の健康・経済的負担:ウガンダにおける実証的研究)

Akampumuza Precious

アカンプムザ プレシャス

Doctoral Thesis

博士論文

The Health-Economic Burden of Climate Change and Climate Variability: An Empirical Study from Uganda

(気候変動・気象変化の健康・経済的負担:ウガンダにおける実証的研究)

Akampumuza Precious

アカンプムザ プレシャス

Graduate Program in Sustainability Science – Global Leadership Initiative

Graduate School of Frontier Sciences

The University of Tokyo

Abstract

The Health-Economic Burden of Climate Change and Climate Variability: An Empirical Study from Uganda

(気候変動・気象変化の健康・経済的負担:ウガンダにおける実証的研究)

Akampumuza Precious

アカンプムザ プレシャス

Supervisor: ONUKI Motoharu

Healthy lives for all is a key pillar of sustainable development as envisioned by the third sustainable development goal (SDG 3). The realization of this goal is however likely to be jeopardized given the current trend of climate change and climate variability. The Intergovernmental Panel on Climate Change projects possible increases in surface temperatures in the range of 2-4°C by 2100. Similarly, heat waves, cyclones, droughts, floods and other extreme weather events have also increased and anticipated to increase further, both in terms of severity and frequency. The potential health impacts of climate change and climate variability are diverse and are both direct and indirect. Indirect effects include but are not limited to injuries, illnesses and deaths due to heat waves, floods and other extreme weather events. Indirect effects include increased incidence of vector borne disease like malaria and dengue fever which result from increased multiplication of disease-causing vectors and/or their expansion into previously inhabitable regions.

Torrential rainfall that results in massive flooding could also be associated with increased spread of water-borne diseases like diarrhea. Malnutrition is another potential indirect effect of climate change and climate variability. Floods, droughts and other extreme weather events

often reduce food production and jeopardize food security, leading to increased malnutrition rates especially among children. The challenge is more pronounced among poor agrarian countries which are heavily reliant on rain-fed agriculture. Despite the increasing scholarly importance of climate-health interlinkages, empirical evidence on the same is scanty and largely concentrated in the Global North while Sub-Saharan Africa is less represented, partly due to data limitations. This is the main motivation for this dissertation, which attempts to provide micro level empirical evidence on the health-economic burden of climate change and climate variability in Uganda.

Uganda has experienced major changes to its traditional climate. Between 1960 and 2010, surface temperatures are estimated to have risen by 1.3°C and are anticipated to rise further by 1.5°C by 2030. Although rainfall totals have not changed substantially, the frequency of torrential rains and unpredictability of rainy seasons pose major threats especially to rural smallholder livelihoods. The number of droughts experienced in some vulnerable regions are particularly worrisome: Between 2001 and 2011, the country experienced five major droughts with devastating consequences. The effect of climate change in general and weather shocks in particular are projected to cause economic damage worth 2-4% of GDP in the four sectors of Agriculture, Water, Energy and Infrastructure. The micro level impacts however remain less investigated especially with regards to public health.

The main objective of this dissertation is to quantify the effects of climate change and climate variability on the incidences of morbidity due to selected climate-sensitive diseases, as well as their economic burden in terms of related healthcare expenditure. The analysis is organized in four main chapters of the dissertation. Chapter 3 presents descriptive analysis of rainfall and temperature trends for Uganda and Teso sub-region for the period 1921-2015. It reveals that although rainfall amounts have not changed substantially during this period, there are clear

changes in the distribution of rainfall across seasons in the past three decades (1980-2010). Specifically, the amount of rainfall is reducing in the first rain season (March-May) and increasing in the first dry season (June-August), possibly indicating either delaying or shifting rainfall from the former to the latter season. Temperature on the other hand increased substantially during the same period especially in the second dry season (December-February). Chapter 4 examines the relationship between climate change and the incidence of diarrhea using outpatient and long-term rainfall and climate data from the Teso sub-region, one of the most vulnerable sub-regions in the country. The study's novelty is the use of outpatient diagnosis data to measure the monthly number of hospital visits linked to diarrhea, as opposed to self-reported illnesses elicited through household surveys which are potentially prone to recall bias. Hypothetically, the number of diarrhea cases is expected to respond to extreme weather, given that the disease is highly sensitive to water availability. Indeed, descriptive analysis confirms this hypothesis: monthly rainfall and hospital visits due to diarrhea are negatively and significantly correlated, indicating higher cases of diarrhea during relatively dry months. Regression results further confirm this negative relationship: Estimates from Poisson, Negative Binomial and OLS regression models show that during months when rainfall is 20% lower than the month-specific long-term (80-year) average, the number of diarrhea cases increases by 11-20%. Analysis disaggregated by age and gender categories indicates that monthly diarrhea cases for males aged 0-4 years, females aged 0-4 years, males aged five years and above and females aged five years and above increase by 13-18%, 11-19%, 17-20% and 12-26%, respectively, when monthly rainfall is 20% lower than the month-specific long-term average. In line with previous literature, and based on focus group discussions during data collection, the pathway of this observed effect seems to be reduced availability of clean water during relatively dry months. This leads people to use unsafe water for drinking and cooking as

well as deterioration in hygiene conditions like hand washing after toilet, ultimately quickening the spread of diarrhea pathogens among community members.

Chapter 5 of the dissertation investigates the quantitative link between drought exposure at the household level and morbidity of household members in Uganda. The main hypothesis in the chapter is that households affected by drought face greater health burden in terms of its members suffering more (self-reported) illnesses. The study utilizes longitudinal survey data from the Uganda National Panel Survey (UNPS) collected by the Uganda Bureau of Statistics in four waves: 2009/10, 2010/11, 2011/12 and 2013/14. Approximately 3,000 households were interviewed in each of the UNPS waves. The Health module elicited information on the illnesses household members suffered in the 30 days before the survey year, which is used in constructing the outcome variables of interest. Marginal effects from Probit regression models reveal members of droughtaffected households are 10-13 percentage points more likely to report illnesses in general. Further results disaggregated by type of illness show that the likelihoods of reporting fever, cough, diarrhea and severe headache are 11-14, 7-8, 5-8 and 12-14 percentage points higher, respectively, among drought-affected households relative to unaffected ones. Albeit significant at only the 10% level, the drought effect on diarrhea is also positive. This indicates that drought exposure deteriorates the health of members of affected households.

In Chapter 6, the economic burden associated with drought exposure among affected households in Uganda are unraveled. Like in chapter 5, analysis in this utilizes UNPS data covering a panel period of four years between 2009 and 2014. Information on household expenditure on healthcare and other consumption components is drawn from the Expenditure section of the household module. This is the source of the main outcome variable, which is the per capita spending on healthcare for the household members that reported illnesses. Basic OLS regression results indicate substantial increase in per capita healthcare expenditure while non-health expenditure significantly reduces when a household is affected by a drought. More robust estimates based on Tobit Random Effects (Tobit-RE) reveal increases in healthcare expenditure in the range of 23-29% while non-health expenditure reduces by approximately 7-10% when a household is affected by a drought. This perhaps indicates that households reallocate resources from non-health consumption to supplement healthcare expenditure during and/after drought.

The results presented in the dissertation carry key implications for the resilience of human health and healthcare systems amidst climate change and climate variability. First, they provide empirical evidence on the micro level impacts of these global threats, which has been critically lacking in Uganda particularly and Sub-Saharan Africa generally. Providing knowledge on this association is a first step to the incorporation of climate knowledge in healthcare planning processes, for example the integrated disease surveillance and response systems. This could help evidence-based strategies to increase the resilience of the health sector, households and individuals. Having reliable information on the climate-health co-dynamics could indeed help in better prediction of disease outbreaks and design appropriate response strategies. Secondly, the finding that diarrhea cases increase during dry months calls for the need to properly plan water resources especially during periods of drought. Thirdly, there is need to raise people's awareness about the potential health effects of droughts, which could help them take precautionary measures including appropriate utilization of water resources, improving hygiene practices, securing precautionary medical supplies and saving funds for the expected increase in healthcare expenditure. Finally, the huge drought-related healthcare cost implies a huge financial burden that could plunge the already poor and vulnerable households into further poverty.

Dedication

I dedicate this dissertation to my parents; Kantobo Loyce and Wamanya Warren, my husband; Kasim Ggombe Munyegera, my three children; Mutebi Abdul-Rahman, Namutebi Lashrah and Mubiru Ramadhan, my brothers and sisters.

Table	of	Contents
	-	

Abstracti
Dedication vi
Acknowledgementsxv
CHAPTER 11
Introduction1
1.1. Background1
1.2. Objective and structure
CHAPTER 2
Conceptual Framework, literature review and study area
2.1. Climate change, climate variability and extreme weather events
2.2. Pathways of climate change influences on human health
2.3. Vulnerability to climate change in the study area – Uganda10
CHAPTER 3
Socio-economic, Health and Climate Change the Study Area13
3.1. Overview of Uganda
3.1.1: Location
3.1.2: Socio-economic status15
3.1.3. Health status
3.1.4: Climate

3.2. Overview of Teso sub-region	
3.2.1. Location	32
3.2.2. Socio-economic status	
3.2.3. Health status	
3.2.4. Climate	34
CHAPTER 4	42
Climate Change and Diarrheal Disease: Evidence from Teso Sub-region of Uganda	42
4.1. Introduction	42
4.2. Data and descriptive analysis	45
4.2.1. Data	45
4.2.2 Descriptive analysis	49
4.3. Estimation	71
4.3.1. Poisson regression model	71
4.3.2. Negative binomial regression model (NBRM)	72
4.3.3. Ordinary least squares (OLS)	73
4.4. Results and discussion	75
4.4.1. Poisson regression model	75
4.4.2. Negative binomial regression model (NBRM)	80
4.4.3. Ordinary least squares (OLS)	82
4.5. Summary	84

CHAPTER 5	85
The short-term health effects of drought in Uganda	85
5.1. Introduction	85
5.2. Data and descriptive analysis	
5.2.1. Data	
5.2.2. Descriptive analysis	87
5.3. Estimation	
5.3.1. Drought exposure and individual level illnesses	
5.3.2. Drought exposure and number of inactive days	
5.4. Results and discussion	
5.4.1. Drought exposure and individual level illness	
5.5. Summary	119
CHAPTER 6	121
The economic burden of drought-induced morbidity	121
6.1. Introduction	121
6.2. Data and descriptive analysis	
6.2.1. Data	
6.3. Estimation	
6.3.1. Drought exposure and per capita healthcare expenditure	135
6.4. Results and discussion	

6.4.1. Drought, per-capita healthcare expenditure and healthcare expenditure ratio
6.5. Summary148
CHAPTER 7
Conclusions and implications
7.1. Conclusion150
7.2. Implications
eferences154
Appendix

List of Figures

Figure 1: Pathways of climate change influences on human health
Figure 2: Map of Uganda showing sub-regions14
Figure 3: Total population of Uganda between 1911 and 201415
Figure 4: Uganda's nominal GDP and GDP per capita in current US dollars 2001-201817
Figure 5: Annual growth rate of Uganda's nominal GDP and GDP per capita 2001-201818
Figure 6: Leading causes of death for under-5 in-patients in 201520
Figure 7: Total annual rainfall for Uganda 1921-2015
Figure 8: Mean monthly temperature for Uganda 1921-201523
Figure 9: 30-year averages of mean monthly rainfall: Uganda 1921-201024
Figure 10: Long-term monthly rainfall for Uganda 1921-201525
Figure 11: Mean monthly rainfall for Uganda 1921 (30-year averages)26
Figure 12: Long-term mean monthly temperature for Uganda 1921-201527
Figure 13: Mean monthly temperature for Uganda 1921-2015 (30-year averages)
Figure 14: Coefficients of variation for Uganda's Rainfall 1921-2015
Figure 15: Coefficients of variation for Uganda's Temperature 1921-2015
Figure 16: Location map of the Teso sub-region
Figure 17: Total annual rainfall for Teso sub-region 1921-2015
Figure 18: 30-year averages of mean monthly rainfall: Teso sub-region 1921-201036
Figure 19: Long-term monthly rainfall for the Teso Sub-region 1921-2015
Figure 20: Monthly rainfall for Teso sub-region 1921-2015 (30-year averages)
Figure 21: Coefficients of variation for Teso sub-region's rainfall 1921-2015
Figure 22: Relationship between rainfall and temperature for Teso sub-region 2006-201140

Figure 23: Monthly rainfall and diarrhea cases in Teso sub-region: 2006-2011	54
Figure 24: Relationship between rainfall and diarrhea by district	56
Figure 25: Relationship between rainfall and diarrhea by year for Teso	58
Figure 26: Relationship between rainfall and diarrhea by year for Kumi	59
Figure 27: Relationship between rainfall and diarrhea by year for Bukedea	60
Figure 28: Relationship between rainfall and diarrhea by year for Kaberamaido	61
Figure 29: Relationship between rainfall category and number of diarrhea cases	62
Figure 30: Relationship between temperature and diarrhea by district	64
Figure 31: Relationship between temperature and diarrhea by year for Teso	65
Figure 32: Relationship between temperature and diarrhea by year for Kumi	66
Figure 33: Relationship between temperature and diarrhea by year Bukedea	67
Figure 34: Relationship between temperature and diarrhea by year for Kaberamaido	68
Figure 35: Drought and the proportion of household members by year	94
Figure 36: Drought and the proportion of HH members reporting illness by age category	96
Figure 37: Individuals reporting illness disaggregated by quintiles of asset value	98
Figure 38: Drought exposure and number of inactive days	100
Figure 39: Number of inactive days disaggregated by quintiles of household asset value	101
Figure 40: Drought and per capita consumption expenditure by year	127
Figure 41: Drought and per capita consumption expenditure by region	128
Figure 42: Drought and per capita healthcare expenditure by year	129
Figure 43: Drought and per capita healthcare expenditure ratio by year	130

List of Tables

Table 1: Expected, missing and available number of observations by district
Table 2: Participants of focus group discussions during the field survey 2017
Table 3: Total number and incidence rate of diarrhea by district and year
Table 4: Number of months with rainfall 20% below long-term average
Table 5: Climate variables 53
Table 6: District characteristics
Table 7: Monthly number of diarrhea cases by rainfall category 70
Table 8: Effect of rainfall and temperature on diarrhea: Poisson regression
Table 9: Effect of rainfall and temperature on diarrhea: NBRM regression
Table 10: Effect of rainfall and temperature on diarrheal cases: OLS regression
Table 11: Composition of individuals across survey years 2010-2014
Table 12: Composition of individuals across survey years 2010-2012
Table 13: Number and proportion of individuals reporting illness 91
Table 14: Number of individuals reporting illness disaggregated by drought and age categories 91
Table 15: Number of individuals reporting illnesses by year and drought exposure status
Table 16: Number and percentage of individuals reporting illness by quintiles of asset value97
Table 17: Individual Level Summary Statistics 103
Table 18: Summary statistics by drought exposure status 104
Table 19: Drought exposure and individual-level self-reported illnesses: Probit model113
Table 20: Drought exposure and individual-level self-reported illnesses: Probit RE model115
Table 21: Drought exposure and the number of inactive days: Tobit models
Table 22: Composition of households across survey years 2010-2014 123

Table 23: Composition of households across survey years 2010-2012	.125
Table 24: Household-level variables by shock exposure status	.134
Table 25: Drought and health expenditure: Tobit RE and Linear FE models	.141
Table 26: Relationship between drought exposure and health expenditure: FE and Tobit RE	.146

Acknowledgements

The success of this dissertation was a result of the combined support of many. Although I can only list a few here, I sincerely appreciate whoever contributed in some way to make it a success. To begin with, I express my sincere gratitude towards the members of my PhD dissertation committee, Professors; (MINO Takashi, ONUKI Motoharu, SUZUKI Aya, KONISHI Shoko, SEKIYAMA Makiko and MATSUDA Hirotaka) for the excellent guidance and constructive comments which were fundamental in shaping this dissertation. My sincere appreciation goes to SEKINE Naomi, OPOKU Yuko and YAMADA Atsuko for the administrative work they have rendered toward me during my stay at the University of Tokyo.

I am heartily grateful to GPSS-GLI and The University of Tokyo for offering me GPSS-GLI Fellowship to fund Masters and PhD studies, without which this dissertation would have been impossible. Special appreciation goes to Graduate School of Frontier Sciences for the research grants offered to me to conduct field survey for extra data collection to accomplish this dissertation. Special thanks to members of our lab; Gideon Baffoe, Yuka Shimamura, Lillian Muasa ,Yangqing Chen, Alexander Sjögren (late), CAI, Yini, and Cecilia Grandi for their time and comments during presentations at the weekly seminars. I would like to extend my gratitude to my husband Ggombe Kasim Munyegera (PhD) for his academic advice and family support especially in taking care of our children without which this work would perhaps be a myth. Above all, I am grateful to the Almighty God for the gift of supportive family, friends and good health during this study.

CHAPTER 1

Introduction

1.1. Background

Climate change and climate variability are ranked among the main threats to global order due to their impending damage to places, sectors of economic development and livelihoods (World Economic Forum, 2018). The manifestations of climate change and climate variability vary from region to region but generally include significant changes in climatological statistics such as average and peak temperatures, humidity, atmospheric pressure, precipitations, wind patterns, and water salinity and decreases in mountain and polar glaciers (McMichael, 2013). Extreme weather events, whose occurrence is reportedly increasing, are one of the major indicators of climate change and climate variability.

The Intergovernmental Panel on Climate Change predicts that surface temperatures could increase by 2-4°C by 2100 (Solomon et al., 2007). Although the causes are quite diverse, there is a consensus in the literature that these are largely human (Griggs and Noguer, 2002; Judkins, 1999). In fact, the single most important cause of global warming is the emission of greenhouse gases from fossil fuel-based power generation, transport, agriculture and mining sectors that increase the heat-retaining capacity of the lower atmosphere (McMichael, 2013; Judkins, 1993). Irrespective of the cause however, what is much daunting is the expected impacts on various sectors and livelihoods, especially in developing countries which contribute the least and have the least adaptive capacity (Patt et al. 2010). Changes in surface temperatures, humidity, atmospheric pressure, precipitations and wind patterns pose an impending threat to environmental, social and economic sustainability. Extreme weather events like floods and droughts are also reportedly on the rise, both in terms of frequency and incidence (IPCC, 2014). Evidence from existing literature

shows that such events are associated with deterioration in human health through different pathways. Human health is among the sectors that are vulnerable to climate change and climate variability (Hear after, the term "health" is used for human health because the latter is the focus of this dissertation). Understanding the co-dynamics of major climate variables and health indicators – for example the incidence of climate-sensitive infectious diseases – is important for healthcare planning amidst climate change and meteorological variability (Ebi et al., 2016). However, the causal relationship between climate change and climate variability on the one hand and health on the other has not been sufficiently investigated empirically, although understanding this relationship is recognized as one of the critical components of climate change adaptation strategy (Bypass, 2009). Particularly, it is predicated that the impact of climate change on health will be most severe in sub-Sahara Africa (Boko et al., 2018).

One of the critical challenges in climate-health research is the difficulty in identifying the complex relationship and to distinguish climate impact on health from the impacts of other surroundings that influence health (Bannister et al., 2015). Existing literature indicates that the potential climate impacts on health are diverse and are both direct and indirect. Direct effects include but are not limited to injuries, illnesses and deaths due to heat waves, floods and other extreme weather events (Haines et al., 2006). Indirect effects include increased incidence of vector borne diseases like malaria and dengue fever which result from increased multiplication of disease-causing vectors and/or their expansion into previously inhabitable regions.

In some highland areas which were less prone to malaria, the incidence of this disease has been shown to increase in response to warmer temperatures which provided conduce breeding conditions for malaria-transmitting mosquitoes (Zhou et el., 2004). Increases in malaria incidences in the Kenyan highlands have also been found to associate with warmer temperatures that provided conducive breading conditions for mosquitoes (Githeko et al., 2004). Torrential rainfall that results in massive flooding could also be associated with increased spread of water-borne diseases like diarrhea (Funari et al., 2012). Malnutrition is also reported as another potential indirect effect of climate change and climate variability. Floods, droughts and other extreme weather events often reduce food production and jeopardize food security, leading to increased malnutrition rates especially among children (Haines et al., 2006; Funari et al., 2012; Sellman and Hamilton, 2007; Vu and Glewwe, 2011). The challenge is more pronounced among poor agrarian communities which are heavily reliant on rain-fed agriculture. Finally, global warming is associated with a higher concentration of pollutants in the atmosphere, which could result in increased cases of respiratory and cardiovascular diseases (Jacob & Winner, 2009).

The growing body of literature on the climate-health interlinkages notwithstanding, empirical studies on the direct and indirect effects of climate change and climate variability are quite limited. Even then, majority of these are focused on the Global North – especially North America and Europe – leaving a big literature gap in the case of the Global South – particularly Sub-Saharan Africa (Verner et al., 2016, Bypass, 2009). The lack of a clear understanding of the relationship between climate change and climate variability on the one hand and human health on the other inhibits strategies to increase the resilience of the health sector and people against these threats in the context of Sub-Saharan Africa. In fact, empirical evidence on the climate-health interlinkage is crucial to understanding the co-dynamics of climate and health indicators and better plan healthcare services to avert and/or cope with the adverse health effects of climate change and climate variability.

1.2.Objective and structure

The main objective of this dissertation is to investigate the health-economic burden of climate change and climate variability in Uganda. Chapter 2 describes the conceptual

framework and reviews existing literature about climate-health interactions. Chapter 3 provides a brief overview of – location, health, climate and socio-economic situation – of two study areas; Uganda and the Teso sub-region of Uganda. Chapter 4 then examines the impact of climate variability on the incidence of diarrheal diseases in the Teso sub-region. In order to analyze the health implications of drought at the national level, chapter 5 investigates the association between drought exposure and morbidity – using self-reported illnesses as a proxy – using nationally representative household survey data. Chapter 6 builds on this analysis by examining the financial implications of drought-induced morbidity estimated as per capita healthcare expenditure attributable to drought exposure. Finally, chapter 7 concludes the dissertation with a summary of the main findings and key recommendations. Although chapters 5 and 6 have a limitation of relying on self-reported illness as in many other previous studies, this chapter benefits from complementary observed outpatient data in chapter 4 which confirms that the relationship between climate change and health exists even after controlling for potential recall bias.

CHAPTER 2

Conceptual Framework, literature review and study area

2.1.Climate change, climate variability and extreme weather events

Climate change is defined differently among literatures. The Inter-party Coalition on Climate Change defines climate change as a change in the state of the climate that can be identified (e.g. using statistical tests) by changes in the mean and/or the variability of its properties, and that persists for an extended period, typically decades or longer (Solomon et al., 2007). In this sense, it refers to any change in climate over time, whether due to natural variability or as a result of human activity. Literally 'climate change' denotes to long-term change in the statistical distribution of weather patterns (e.g. temperature, precipitation, etc.) over decades to millions of years of time (Rahman, 2013). On the other hand, the United Nations Framework Convention on Climate Change (UNFCCC) defines climate change as "a change of climate which is attributed directly or indirectly to human activity that alters the composition of the global atmosphere and which is in addition to natural climate variability observed over comparable time periods" (UNFCC, 1994). The Intergovernmental Panel on Climate Change (IPCC) provides an alternative definition of climate change, as the occurrence of a value of a weather variable above (or below) a threshold value near the upper (or lower) ends of the range of its observed values in a specific region (IPCC, 2011). This definition of climate change includes change due to natural variability alongside human activity (UNFCC, 2011).

According to Wu et al. (2016), climate change refers to long-term statistical shifts of the weather, including changes in the average weather condition or in the distribution of weather

conditions around the average, that is, extreme weather events. Climate variability on the other hand refers to variations in the mean state of climatological statistics on all temporal and spatial scales beyond those of individual weather events (UNFCC, 2011). Variability may result from natural internal processes within the climate system (internal variability) or from variations in natural or anthropogenic external forces (external variability) (Smith, 2010). Smith et al (2010) provides criteria to differentiate climate variability from climate change. According to their definition, climate variability takes place over a shorter period, from months to less than 30 years while the climate change occurs over a time horizon longer than 30 years (Smith et al., 2010). climatological statistics

As above mentioned, climate change defined as long-term shifts of the weather from climatological statistics by Wu et al. (2016) includes extreme weather events which is referred as changes in the average weather condition or in the distribution of weather conditions around the average. These are defined as events that have extreme values of certain meteorological variables (Stephenson et al., 2008). Typical examples of extreme weather events are excessive rain that often results in floods, droughts, floods, storms, among others (McMiacheal et al., 2004; Karl et al., 1995). This dissertation define extreme weather events as months during which rainfall and temperature values are unusually below or above the month-specific long-term average. Extreme weather events are linked to the ongoing trend of both climate change and climate variability in such a way that both the frequency and intensity of extreme weather events like floods and droughts increase as both climate change and climate variability take place.

2.2.Pathways of climate change influences on human health

Climate change and climate variability and the associated extreme weather events influence human health through several pathways which are often exacerbated and amplified by other global changes. The 1948 Constitution of the World Health Organization defines health as a state of complete physical, mental and social wellbeing which is not simply the absence of disease or illness (WHO, 2006). Other scholars give an alternative approach to the definition of health as personal satisfaction level of individuals in society (Bellieni and Bounocore, 2009). In this dissertation, health refers to the state of wellness in which individuals are free from disease or illness. As summarized in Figure 1, climate change works through both direct and indirect pathways to influence human health. Climatic conditions influence environmental systems and social conditions which affect food yields, water supplies and disease patterns and ultimately pose risks on human health. Direct health effects include injury, death and post-event traumatic stress. The indirect effects are of three broad categories. The first category is the disruption of food production systems and food price hikes that result in malnutrition and other diseases related to food insufficiency and nutrition deficiency. The second category is the deterioration of water resources which leads to increased spread of water-borne diseases like diarrhea. The final category is the alteration of microbial ecology, leading to increased spread of malaria, dengue fever and other vector-borne diseases.



Figure 1: Pathways of climate change influences on human health

Source: McMicheal (2013).

Existing literature shows that the health effects of climate change and climate variability could be both positive and negative, partly depending on geographical location. Mid-latitudes are anticipated to benefit from reduced pneumonia, bronchitis and arthritis following warmer surface temperatures (Franchini & Mannucci, 2015). In many parts of the world however, the observed and projected health effects are largely negative, especially those resulting from increasing surface temperatures. The main observed effects of higher temperatures include higher cases of acute coronary syndromes and myocardial infarction (Dilaveris et al., 2006; Goerre et al., 2007), heat-related illnesses and deaths among children and the elderly in (Medina-Ramon et al., 2006), hospital visits related to angina pectoris (Abrignani et al., 2009) and asthma (Lin et al., 2009).

Drought have also been linked to higher cases of diarrheal diseases in 13 African countries (Bandyopadhayay et al., 2012). The reported pathway is the reduced availability of clean water and the deterioration of hygiene practices during extremely dry seasons. In addition, long-term health effects of water stress and droughts have been empirically identified. The study however was restricted to diarrhea and made no reference to other potentially climate-sensitive illnesses. It is important to understand how climate-sensitive diseases respond to changing and varying climatic conditions. Another study that attempted to analyze the health effects of climate variability reported potential positive association between rainfall on the one hand and malaria and gastrointestinal illnesses on the other (Labbé et al., 2016). However, the study used a qualitative approach rather than quantification of the malaria-rainfall relationship. Additionally, it used information only from the Southwestern region and as such, the findings therein are not country representative. The growth rate of rural Zimbabwean children aged 12-24 months is retarded by 1.5-2cm when subjected to rainfall 20-30% lower than the long-term average. By symmetry, rainfall figures significantly higher than long-term averages are associated with higher survival rates for Indian girls relative to boys

(Rose, 1999). In Vietnam, Lohman and Lechtenfield (2015) provide empirical evidence on the shortterm negative impacts of drought exposure on the health condition of household members and the associated increase in healthcare expenditure.

2.3.Vulnerability to climate change in the study area – Uganda

Climate change and climate variability are likely to exacerbate the health burden in Uganda, a country that never achieved most of the health-related millennium development goals (MDGs) especially on child and maternal mortality (WHO, 2018). The country has experienced enormous changes to its climate, including increases in surface temperatures by 1.5°C between 1960 and 2010 (Future Climate for Africa, 2011). With increased occurrence and intensity of droughts and floods, the socio-economic danger is imminent. It is estimated that the four sectors of agriculture, energy, water and infrastructure, the economic loss resulting from climate change and climate variability is projected to amount to 2-4% of cumulated GDP between 2010 and 2050 (Uganda Ministry of Water and Environment, 2015). The potential threat to the agriculture sector is particularly daunting. Between 2006 and 2013, two thirds of Ugandan households that graduated from poverty were reported to move back into it, partly as a result of weather shocks (World Bank, 2016). Government spending on public transfers is also too low to insulate the bottom poor from poverty and vulnerability: public transfers in Uganda account for 0.4% of GDP, a very low proportion relative to 1.1% Sub-Saharan Africa average (World Bank, 2016).

The 2016 World Bank Poverty Assessment Report for Uganda provides preliminary empirical evidence on the pathways through which climate variability – with a focus on rainfall shortages – translate into socio-economic outcomes. The estimates from household survey data between 2006 and 2013 reveal that a 10% rainfall shortage results into significant reductions in agricultural incomes, which range between 38% to 8.7% in the Northern and Eastern regions,

respectively. Likewise, reduction in water sufficiency by 10% comes plummets household per capita consumption by 4.8%. Akampumuza and Matsuda (2017) corroborate these findings by demonstrating that household consumption expenditure per adult equivalent increases by 17% following drought exposure in the Teso sub-region.

Approximately half of the country's disease burden is constituted by communicable diseases including climate-sensitive diseases like diarrhea and respiratory infections (WHO, 2018). There are some clear indications on the possible link between climate change and variability on the one hand and human health on the other. The 2017 global Climate Risk Index report places Uganda in 57th position out of 191 countries, with an annual average of 33 fatalities linked to climate change over the past two decades between 1996 and 2015 (Kreft et al., 2017). This is clearly indicative of the fact that individuals and households vulnerable to climate change and climate variability could face dramatic deterioration to their health status. Water stress following long dry spells could spur the spread of water-borne diseases.

Understanding the patterns and causes of diseases is crucial predicting and planning healthcare services to counter major climate-related epidemics (Lohman and Lechtenfeld, 2015). It is however intriguing that Uganda has not initiated actions to build institutional and technical capacities to work on climate change as reported in the 2015 WHO/UNFCC Climate and Health Country Profile for the country. Likewise, a national assessment on the health impacts of climate change is largely inexistent in the country. The healthcare budgeting processes have also not allocated sufficient funds to strengthen the resilience of the health infrastructure, neither has climate information been incorporated in the Integrated Disease Surveillance and Response (IDSR) system. This dissertation attempts to generate knowledge and empirical evidence on this under-researched topic with a view of informing health policy to increase the resilience of the health sector and the population against adverse climate and weather extremes. The analysis in dissertation uses a combination of disease diagnosis data from outpatient records, longitudinal household survey data and climate records.

CHAPTER 3

Socio-economic, Health and Climate Change the Study Area

This dissertation is based on two main study areas, that is, Uganda and Teso sub-region. The role of this chapter is to provide an overview of the climatic, health and socio-economic conditions of the two study areas.

3.1. Overview of Uganda

3.1.1: Location

Uganda is located in East Africa, with common borders with Kenya in the East, South Sudan in the North, Democratic Republic of Congo in the West and Tanzania and Rwanda in the South. The country is composed of four administrative regions – Central, Western, Eastern and Northern. These are then divided into sub-regions – West Nile, Acholi, Karamoja and Lango from the Northern region; Western and Southwestern from Western region; Teso and Elgon from Eastern region and Central 1 and Central 2 from Central region. The sub-regions are presented in Figure 2. Sub-regions are then sub-divided into 121 districts and 146 counties. The counties are further sub-divided into sub-counties, parishes and villages, which are the lowest units of administration.



Figure 2: Map of Uganda showing sub-regions

Source: Made by author

Note: GIS shapefiles were obtained from openAfrica.

3.1.2: Socio-economic status

Figure 3 presents the trend of Uganda's population recorded every after 10 years between 1911 and 2014, as presented in the 2017 Statistical Abstract (UBOS, 2017). The country's population is on a positive trajectory, increasing exponentially from decade to decade. With annual population growth rate estimated at 3.3% in 2017, the country has among the fastest growing populations in the world (World Bank, 2017).



Figure 3: Total population of Uganda between 1911 and 2014

Source: Uganda Bureau of Statistics Statistical Abstract 2017 based on census data.

Uganda experienced annual growth rate of gross domestic product (GDP) of four percent in 2017 (World Bank, 2018). Indeed, average growth rate was 4.5% for the past five years, higher than the average growth rate recorded in Su-Saharan Africa in the same period. GDP per capita however increased only slightly from to reach USD 621 in the same year (World Bank, 2018). The 2016 World Bank Economic Outlook Report for Uganda estimated that 33.2% of households in the country (representing 43% of the population) lived below the international poverty line of \$1.9 per day in 2015, which is an improvement from 41.5% in 2010 (World Bank Economic Outlook 2016).

Figure 4 shows Uganda's gross domestic product (GDP) and GDP per capita for the period 2001-2018. The two variables exhibit a positive trend over this period. GDP was approximately six billion United States dollars in 2001 and by 2018, it had increased to approximately 28 billion US dollars. Similarly, GDP per capita increased from 240 US dollars in 2001 to 643 US dollars in 2018. In order to examine the trend of the country's economic growth over the same period, Figure 5 presents the growth rate of Uganda's GDP and GDP per capita. Both variables fluctuated during this period with a continued decline in both variables for period between 2014 and 2017 and then increased in 2018. According to the 2016/17 Uganda National Household Survey (UNHS), the proportion of Ugandans living below the national poverty line was estimated at 27% in 2017 (UBOS, 2018). It is a remarkable improvement from 34.6% in 2013 and 53.2% in 2006.

Agriculture, which is the main source of income and livelihood, employs approximately 25.6 million people (68% of the country's population) according to the 2016/17 Labor Force Survey (UBOS, 2018). The sector is predominantly rain fed (Thomas et al., 2007; Thornton et al., 2009) and smallholder with 90% of farms less than two hectares (Lowder et al., 2016).



Figure 4: Uganda's nominal GDP and GDP per capita in current US dollars 2001-2018 Data source: World Development Indicators.



Figure 5: Annual growth rate of Uganda's nominal GDP and GDP per capita 2001-2018 Source: Made by author using data from World Development Indicators.

Climate change and the associated extreme weather events could erode the potential gains from the recently realized economic growth. At the macro level, the damage from climate change to agriculture, water, infrastructure and energy sectors is estimated at 2-4% of GDP between 2010 and 2050 (Uganda Ministry of Water and Environment, 2015). At the micro level, vulnerability to extreme weather events pose further danger of throwing people back into poverty. The threat is amplified by the fact that over two thirds of the population derive their livelihoods from agriculture, the biggest percentage of which is rain-fed and subsistence. Since extreme weather events disproportionately affect the welfare and livelihoods of the poor (Mendelsohn et al., 2006; Bohle et

al., 1994), their increased frequency and intensity of occurrence could pose serious implications regarding income distribution in the country. The country is already facing daunting inequality problem with a GINI coefficient of 0.395 in 2013 (UNDP, 2015). Inequality has already evidently discounted development outcomes in the country: adjusting for inequality reduced the country's Human Development Index (HDI) from 0.463 to 0.274 in 2013 (UNDP, 2015).

3.1.3. Health status

Uganda's health sector registered improvement in a number of health indicators between the financial years 2015/16 and 2016/17. Access to antenatal care was reported to increase by 11% between the two periods while supervised deliveries and postnatal care attendance improved by 26%, respectively (WHO, 2018). Nonetheless, the sector still faces a number of challenges especially with regards to increasing and persistent trends of certain infectious diseases. Physical access to health facilities remains a challenge: urban and rural residents travelled 2.0km and 3.2km, respectively, to access public hospitals in 2016. As of 2015, life expectancy at birth among men and women was estimated at 60 and 65 years, respectively (Ministry of Health, 2015). Maternal and under-five mortality rates remained as high as 336 and 64 per 1,000 people in 2017, respectively (Ministry of Health, 2016). Figure 6 presents the major causes of death among under-five in-patients for the financial year 2015/16, based on the Ministry of Health's Annual Health Sector Performance Report for Financial Year 2015/16 (Ministry of Health, 2016). The figure reveals that 5,295 (43%) of under-five in-patient deaths registered were caused by malaria. Other major causes of death included pneumonia, anemia, neonatal sepsis (for newborns aged 0-7 days), diarrhea, septicemia, injuries (trauma due to other causes), respiratory infections, injuries due to road traffic accidents and gastrointestinal disorders. Malnutrition is yet another key health challenge in the country, as 33.5% of children under five years of age were stunted in 2011 (Mawa and Lukwago, 2018). The status of hygiene practices in the country is also quite poor,
with approximately 25% and 78% of households having no access to a toilet and hand washing facility, respectively by 2016 (Ministry of Health, 2016). This could indeed exacerbate the spread of hygiene-related infectious diseases like diarrhea.



Figure 6: Leading causes of death for under-5 in-patients in 2015 Source: Uganda Ministry of Health (2016).

3.1.4: Climate

The climate of Uganda is bimodal in nature, with two rainfall and dry seasons (Egeru, 2012). The first and longer rainy season lasts between March and May while the second and shorter rainy season occurs between October and November (Mubiru et al., 2012). The first and shorter dry season occurs between June and August, while the second and longer dry season lasts between December and February (Nimusiima et al., 2013). There are however claims that the country's climate is gradually changing and becoming increasingly marred with extreme weather events like droughts and floods whose occurrence is becoming not only more frequent but also more intensive (Irish Aid, 2017). In the decade between 2001 and 2011, the country experienced five major droughts – in 2001, 2002, 2005, 2008 and 2011 (Oxfam, 2008; Guha-Sapir et al., 2012).

In this section, Uganda's climate and rainfall are analyzed using climate data covering the period 1921-2010, obtained from the Climate Change Knowledge Portal (2019). For most of the time in the section, rainfall and temperature variables are analyzed in 30-year periods. The choice of 30-year periods is based on the fact that the World Meteorological Organization recommends using at least 30-year periods for deriving climatological values (Brisson et al., 2015). Indeed, three decades constitute the classical period for performing the statistics used to define climate (Goosse et al., 2010). Other scholars have projected that surface temperatures could rise further by 1.5°C by 2030 while the number of extremely hot days could rise by 15-43% by 2060 (Future Climate for Africa, 2011).

Figures 7 and 8 show total annual rainfall and mean monthly temperature for Uganda for the period 1921-2015. Although there does not seem to be a systematic increase or decrease over the years, rainfall is volatile as it fluctuates from year to year. Temperature seems to be on a positive trajectory especially in the past two decades between 1990 and 2010.



Figure 7: Total annual rainfall for Uganda 1921-2015

Source: World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/



Figure 8: Mean monthly temperature for Uganda 1921-2015

In Figure 9, the same rainfall values are presented for three time periods; 1921-1950, 1951-1980 and 1981-2010. From one 30-year period to another, there are some changes in total annual rainfall. An increase of 76mm (6.6%) was observed between the periods 1921-1950 and 1951-1980 followed by a decline of 14mm (1.1%) between 1951-1980 and 1981-2010.

Source: World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/



Figure 9: 30-year averages of mean monthly rainfall: Uganda 1921-2010

Figure 10 presents monthly rainfall for Uganda for the period 1921-2015 disaggregated by month. Figure 11 then presents 30-year averages of monthly rainfall between 1921 and 2015. Both figures reveal that while rainfall has increased in some months like January, September, October and November, others like February, April and May have exhibited slight decrease in rainfall. Other months like March, April, June, July, August and December have shown no change in rainfall during the analysis period. The clearest changes happened in the second rainy season between September and November. Rainfall in this season increased systematically between 1921 and 2015. There is also a slight decrease in April and May rainfall and a slight increase in June rainfall. A decrease in April and May and an increase in June could perhaps indicate that rainfall towards the first rain season (April and May) could be delaying and instead falling at the start of the first dry season in June. There is no systematic change in rainfall in the second dry season between December and February (Mubiru et al., 2012).

Source: World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/



Figure 10: Long-term monthly rainfall for Uganda 1921-2015

Source: World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/

There are some observable spikes in the rainfall trends for some months. For instance, February rainfall for the year 2010 was 142.1mm, much higher than the long-term average of 53.2mm for the entire period 1921-2010. For the case of 2010, this observation is consistent with official reports that the various parts of the country experienced unusually heavy rainfall that resulted in flooding particularly in the East (IFRCRCS, 2010). Similarly, 280.1mm of rainfall were recorded in September 2007, quite higher than the long-term average of 115.7mm. This confirms official reports of major floods that affected many parts of the country (FAO and WFP, 2008). Likewise, rainfall in September 2012 was 216.5mm which was higher than the long-term September average. This coincides with reports of torrential rainfall that caused severe flooding especially in



the North (ReliefWeb, 2012). Other cases of observable rainfall spikes include June 2015, June 2012, June 2010 and November 2015.

Figure 11: Mean monthly rainfall for Uganda 1921 (30-year averages)

Source: World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/

The trend of mean monthly temperature is presented in Figure 12 for the period 1921-2015. Figure 13 then presents mean monthly temperature disaggregated by three 30-year periods; 1921-1950, 1951-1980 and 1981-2010. Clearly, mean monthly temperature has increased over the analysis period for almost all the 12 months of the year. For most of the months, the increase seems to be more pronounced in the past three decades between early 1990's and mid 2010's. The first dry season (June-August) seems to have become hotter, as temperatures for the months of June, July and August increased between 1921 and 2015. On the other hand, the second rainy season (September-November) exhibited no notable change in temperature during the same period. The second dry season also showed a positive temperature trend, with figures increasing in December, January and February. Besides the increasing trend in these months, there are observable spikes in some years, representing higher temperatures compared to the long-term average recorded in the 1921-2010 period. In December 2003, average monthly temperature was 25.8°C, which is much higher than the average of 22.8°C recorded in the 1921-2010. Similarly, February temperature was 26.6°C in 2008, much higher than that recorded in the period 1921-2010 which was 23.4°C. With 26.5°C, January 2004 was also a hot month relative to the long-term average.



Figure 12: Long-term mean monthly temperature for Uganda 1921-2015

Source: World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/



Figure 13: Mean monthly temperature for Uganda 1921-2015 (30-year averages)

Figures 14 and 15 present the coefficients of variation (CVs) for Uganda's rainfall and temperature, respectively, for the period 1921-2015. The CV value for each time period is calculated as the ratio of the standard deviation of the respective variable to its mean. This helps to show how much the value of the respective variable in each time period is dispersed around the mean. In other words, the higher the CV, the higher the variable is dispersed around the mean and vice versa. The advantage of using CV is that it enables comparison of dispersion of two or more variables which are measured in different scales, for example rainfall (millimeters) and temperature (degrees Celsius).

Source: World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/

For instance, the CVs for rainfall and temperature for Uganda were 47.1% and 5.1%, respectively, for the period 1986-2015. Clearly, his indicates that rainfall was more dispersed that temperature during this time period. In both figures, CVs for rainfall and temperature are plotted against time periods, which are divided into 30-year moving intervals. For instance, the first period represents 1921-1950; the second period represents 1922-1951, and so on.

The figures show increased variation of rainfall in February, April, June, July and August. The highest increase in rainfall variation occurred in September. This, and the previous findings of increasing rainfall in September, could indicate that the month could be experiencing higher and torrential rainfall between 1921 and 2015. On the other hand, the dispersion of November rainfall reduces over time, reduced dispersion of rainfall in this month. With regards to temperature, there is a positive trend in the coefficient of variation for almost all months, indication increasing dispersion of temperature especially during the most recent 30 periods (between 1957-1986 and 1986-2015).



Figure 14: Coefficients of variation for Uganda's Rainfall 1921-2015

Source: World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/



Figure 15: Coefficients of variation for Uganda's Temperature 1921-2015

Source: World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/

3.2. Overview of Teso sub-region

3.2.1. Location

Teso sub-region is located in the Eastern region of Uganda as presented in Figure 16. It comprises of eight districts – Kumi, Ngora, Soroti, Serere, Amuria, Bukedea, Kaberamaido and Katakwi. With a total land area of 13,027 square kilometers and total population of 1,819,790 people, the sub-region has an average population density of 140 per square kilometer of land, based on the 2014 population and housing census (UBOS, 2014).



Figure 16: Location map of the Teso sub-region

Source: Made by author

Note: GIS shapefiles were obtained from openAfrica.

3.2.2. Socio-economic status

The sub-region also lags behind many other sub-regions regarding socio-economic status, particularly poverty levels. The Eastern region – where the Teso sub-region is located – had the highest poverty rate in Uganda in 2018, estimated at 24.5% which is much higher than the national average of 19.7% (World Bank, 2018). Within the Eastern region, Teso sub-region has the highest poverty rate estimated at approximately 28%. Farming is the main economic activity, and major crops grown include cassava, sorghum, millet sweet potatoes and ground nuts on a subsistence scale. Farming in the sub-region is predominantly smallholder, practiced on farms of three acres in size on average and is heavily reliant on natural conditions – particularly rainfall – as irrigation systems are not readily available (Uganda Investment Authority, 2016). This poses a great risk to crop yields and farm incomes.

3.2.3. Health status

Like the national level situation presented earlier, health in the Teso sub-region is marred with numerous challenges related to the prevalence of certain diseases, physical access to and quality of healthcare services, among others. According to the 2014 Demographic and Health Survey, 14% of the sub-region's children were stunted (UBOS, 2016). The same report estimated that the proportion of households with access to an insecticide-treated mosquito net was only 64%, which presents a high risk of malaria in the area. There is a high prevalence of anemia (hemoglobin levels below 110g/dl); 59% of children aged 6-59 months and women aged 15-49 years were characterized as anemic in 2014 (UBOS, 2016). Approximately 37% of the population in the sub-region travel more than a kilometer to access safe drinking water, which compounds the risk of diarrhea due to the use of unsafe water. This proportion is much higher than that reported at the national level in the same year, which was approximately 21%.

3.2.4. Climate

With regards to climate, Teso enjoys a bimodal type of climate similar to the country-level type described in the previous section. The traditional climate however seems to undergo gradual changes over the past decades. Figure 17 reports annual rainfall for the sub-region between and 2015. Overall, the sub-region's rainfall fluctuated over the years, with some indication of slight decreases in rainfall over the past two decades.



Figure 17: Total annual rainfall for Teso sub-region 1921-2015 Source: Made by author using rainfall data from the Uganda National Meteorological Authority.

Similar rainfall trends for the Teso sub-region are presented in Figure 18, in which the analysis period is segregated into three 30-year categories During the periods 1921-1950 and 1951-1980, the sub-region experienced an increase in rainfall by 62.4mm (4.6%). This was however followed by a decline of 33.9mm (2.4%) between the periods 1951-1980 and 1981-2010. This decline is much higher than that presented earlier for Uganda. Precisely, annual rainfall reduced by 15% between 1951-1980 and 1981-2010. It is therefore possible that the pace of climate change in Uganda could vary by geographical area, with Teso sub-region likely to experience greater changes than the national average. Indeed, the sub-region has been cited among the most vulnerable areas to climate change and climate variability in the country (Egeru, 2012). For instance, it experienced five droughts between 1990 and 2010 - 1998, 1999, 2002, 2005 and 2008. During the 2007 flood that swept through Eastern and the Horn of Africa in 2007, about 30 people in the sub-region died and an estimated 8,500 acres of cropland was lost. This was followed by a drought in 2008 which was associated with severe hunger. Shortly after, the sub-region was struck by a drought in 2009, followed by a short drought and flood in 2010 (FAO/WFP, 2008). Reports from the district agriculture offices confirmed tremendous declines in food production following periods of massive floods and droughts (FAO/WFP, 2008). This is often a combined result of crop loss in producing villages and disrupted food flow channels when roads are destroyed by floods. Micro level evidence indicates that when households in the sub-region are exposed to drought, they experience significant declines in food security and that the decline is more pronounced among female-headed households who have less coping capacity (Akampumuza and Matsuda, 2017).



Figure 18: 30-year averages of mean monthly rainfall: Teso sub-region 1921-2010 Source: Made by author using rainfall data from the Uganda National Meteorological Authority.

Figure 19 below presents monthly rainfall for the Teso sub-region over between 1921 and 2015. As in the case of Uganda presented earlier, there are spikes in rainfall in some of and years. For example, monthly rainfall for February 2010 was 212.6mm which was quite higher than the long-term average of 58.9mm recorded in the period 1921-2010. Other months with high rainfall in May include 427.2mm and 259.3mm recorded in 1991 and 1998, respectively. In November, rainfall spikes are observable in 1972 and 2008. Generally, the past three decades between 1980 and 2010 have more spikes, indicating increasing rainfall variability in the sub-region.



Figure 19: Long-term monthly rainfall for the Teso Sub-region 1921-2015 Source: Made by author using rainfall data from the Uganda National Meteorological Authority.

Similar trends of monthly rainfall are then presented in Figure 20, disaggregated not only by month but also by three 30-year periods. While there is a clear decrease in April and May rainfall between 1951-1980 and 1981-2010, October rainfall slightly increased in the same period.



Figure 20: Monthly rainfall for Teso sub-region 1921-2015 (30-year averages) Source: Made by author using rainfall data from the Uganda National Meteorological Authority.

Coefficients of variation for monthly rainfall for Teso sub-region are presented in Figure 21. A decreasing trend for the coefficients of variation in January suggests that rainfall in this month has become less variable over the analysis period. On the other hand, the trend for May is positive, indicating increasing variability of May and October rainfall between 1921 and 2010.



Figure 21: Coefficients of variation for Teso sub-region's rainfall 1921-2015

Source: Made by author using rainfall data from the Uganda National Meteorological Authority.

Figure 22 plots the monthly rainfall, minimum and maximum temperatures for Teso sub-region for the period 2006-2011. The relationship between rainfall and maximum temperature is negative, suggesting that months with low rainfall coincide with high temperature and vice versa.



Figure 22: Relationship between rainfall and temperature for Teso sub-region 2006-2011 Source: Made by author using rainfall data from the Uganda National Meteorological Authority.

3.3. Summary

This chapter has presented an overview of the climate and socio-economic and health situation of Uganda and Teso sub-region. The chapter reveals that the bimodal type of Uganda's rainfall has not changed over the period 1921-2015. However, the distribution of rainfall across months and seasons has changed during the same period. Rainfall in the months of April and May reduced over the past three decades while that of June increased. This perhaps indicates that the first rain season rainfall is increasingly delayed and instead experienced at the beginning of the first dry season. On the other hand, the country's temperature increased between 1980 and 2010, particularly in the second dry season between December and February. The chapter also reveals a connection between climate change and climate variability and socio-economic outcomes. This suggests the possibility that climate change and climate variability could exacerbate the already high rates of poverty, vulnerability and inequality in a country.

Chapter _4___ (pp. __42__ to __84___) of my doctoral thesis cannot be made public on the Internet for (5) years from the date of doctoral degree conferral because that part will be submitted for publication.

Chapter _5___ (pp. _85___ to __109__) of my doctoral thesis cannot be made public on the Internet for (5) years from the date of doctoral degree conferral because that part will be published.

Chapter _6___ (pp. _121___ to __149__) of my doctoral thesis cannot be made public on the Internet for (5) years from the date of doctoral degree conferral because that part will be published.

CHAPTER 7

Conclusions and implications

7.1. Conclusion

Climate change and climate variability are among the greatest threats to global stability and order due to their impending danger on various sectors. With regards to human health, their observed and projected effects are both direct and indirect. The direct pathways include but are not limited to heatwaves associated with respiratory infections and heat-related mortality. Examples of indirect pathways include alteration of disease patterns, resurgence and emergence of existing diseases and emergence of new ones especially in previously uninfected regions. Empirical evidence on the climate-health linkage is however still scanty especially in developing countries, partly due to data unavailability. This dissertation contributes to closing this research gap by empirically examining the health-economic burden of climate change and climate variability in Uganda, a Sub-Saharan African country with limited empirical evidence on the topic.

The analysis is presented in four parts. The first part (chapter 3) provides a detailed descriptive analysis of rainfall and temperature trends for the period 1921-2015. The chapter reveals that notable changes in rainfall and temperature especially in the past three decades between 1980 and 2010. Rainfall in the first rainy season (March-May) reduced while that of June increased, possibly indicating shifting rainfall season from the first rain season to the first dry season. With regards to temperature, the second dry season (December-February) seems to have become hotter during the same period. The second part (chapter 4) analyzes the effect of abnormally low rainfall on the incidence of diarrhea using outpatient and climate data from Teso sub-region of Uganda, a region highly susceptible to weather shocks. The key novelty of this study is the use of actual disease diagnosis from outpatient records as a measure of observed diarrhea cases as opposed to

reliance on self-reported illnesses from household surveys. This ameliorates potential recall bias in reporting exposure to diarrheal cases. Regression results indicate that outpatient visits related to acute diarrhea tend to increase during months when rainfall figures are 20% lower than the month-specific 80-year average. It is presumed that the pathway to the observed effect is the reduced availability of safe drinking water during extremely dry months which presents unprotected springs and ponds as inevitable alternative sources.

It is important to note that the above results represent a single region of Uganda; Teso subregion which has been selected as the appropriate area to establish climate change evidence. Given the fact that the whole country is in amidst of public health crisis due to climate change, it is therefore vital to conduct a nationally representative investigation of the relationship between climate change and climate variability on the one hand and morbidity on the other. The third part (chapter 5) of the dissertation analyzes the relationship between household level exposure to drought and the incidence of illnesses at the individual level using nationally representative household survey data from the Uganda National Panel Survey (UNPS). The results reveal that household members subjected to drought are more likely to report illnesses in general and fever, cough diarrhea and severe headache in particular.

From the economic burden point of view, different reports and published articles have acknowledged the fact that climate change will cause huge financial losses at regional, national and household level. It is expected that the poor – whose ability to cope with the consequences of climate change is limited – will face increased financial burden in the event that weather shocks strike. However, empirical evidence to support the claim is still weak. Therefore, the fourth and final part (chapter 6) of the dissertation explores the financial burden of drought-related morbidity using four waves of the Uganda National Panel survey conducted by the Uganda Bureau of

Statistics between 2009 and 2014. Overall, results indicate that households that report having experienced drought within 12 months of the survey date spend 23-29% more on healthcare relative to those unaffected by drought. Similarly, drought-affected households spend a larger proportion of their budgets on healthcare and less on non-healthcare expenditure, partly supporting the conjecture of resource reallocation away from the latter to supplement the former during and/or after drought.

7.2. Implications

Several implications emerge from the results presented in this dissertation. First, they provide empirical evidence on the micro level impacts of climate change and climate variability. In particular, empirical evidence on the impact of drought on health has been critically lacking in Uganda particularly and Sub-Saharan Africa generally. Providing knowledge on this association is a first step to the incorporation of climate knowledge in healthcare planning processes, including the integrated disease surveillance and response systems. This could help increase the resilience of the health sector, households and individuals to the potential adverse effects of climate change and climate variability. Adequate and precise knowledge on the co-dynamics of climate variables and climate-sensitive infectious diseases could indeed help in better prediction of disease outbreaks and design appropriate response strategies to counter them. Secondly, the finding that diarrhea cases increase during dry months calls for the need to properly plan water resources especially during periods of extreme weather. This also entails sensitization of people to not appropriately manage scarce water resources and improve hygiene practices even during episodes of droughts and water shortage. There is also need to create people's awareness to be on the lookout for certain infectious diseases when droughts are anticipated. This could help households to better prepare for weather shocks by securing precautionary medical supplies and saving funds for the expected increase in

healthcare expenditure. Finally, the increase in out-of-pocket healthcare expenditure by droughtaffected households implies that such weather shocks could plunge the already poor and vulnerable households into further poverty.

References

- Abrignani, M. G., Corrao, S., Biondo, G. B., Renda, N., Braschi, A., Novo, G. & Novo, S. (2009). Influence of climatic variables on acute myocardial infarction hospital admissions. *International journal of cardiology*, 137(2), 123-129.
- Ahern, M., Kovats, R. S., Wilkinson, P., Few, R., & Matthies, F. (2005). Global health impacts of floods: epidemiologic evidence. *Epidemiologic reviews*, 27(1), 36-46.
- Akampumuza, P., & Matsuda, H. (2017). Weather shocks and urban livelihood strategies: The gender dimension of household vulnerability in the Kumi District of Uganda. *The Journal* of Development Studies, 53(6), 953-970.
- Alexander, K., Carzolio, M., Goodin, D., & Vance, E. (2013). Climate change is likely to worsen the public health threat of diarrheal disease in Botswana. *International journal of environmental research and public health*, *10*(*4*), *1202-1230*.
- Anderson, G. B., Dominici, F., Wang, Y., McCormack, M. C., Bell, M. L., & Peng, R. D. (2013). Heat-related emergency hospitalizations for respiratory diseases in the Medicare population. *American journal of respiratory and critical care medicine*, 187(10), 1098-1103.
- Asfaw, S., & Maggio, G. (2017). Gender, weather shocks and welfare: evidence from Malawi. *The Journal of Development Studies*, 1-21.
- Bandyopadhyay, S., Kanji, S., & Wang, L. (2012). The impact of rainfall and temperature variation on diarrheal prevalence in Sub-Saharan Africa. *Applied Geography*, *33*, 63-72.
- Bartlett, M. S. (1937). Properties of sufficiency and statistical tests. *Proceedings of the Royal Society of London. Series A-Mathematical and Physical Sciences*, *160*(901), 268-282.

- Bastos, P., Busso, M., & Miller, S. (2013). Adapting to climate change: Long-term effects of drought on local labor markets.
- Beegle, K., Dehejia, R. H., & Gatti, R. (2006). Child labor and agricultural shocks. Journal of Development economics, 81(1), 80-96.
- Bell, M. L., Goldberg, R., Hogrefe, C., Kinney, P. L., Knowlton, K., Lynn, B., ... & Patz, J. A. (2007). Climate change, ambient ozone, and health in 50 US cities. *Climatic Change*, 82(1-2), 61-76.
- Bellieni, C. V., & Buonocore, G. (2009). Pleasing desires or pleasing wishes? A new approach to health definition. *Ethics & Medicine*, 25(1), 7-12.
- Bohle, H. G., Downing, T. E., & Watts, M. J. (1994). Climate change and social vulnerability: toward a sociology and geography of food insecurity. *Global environmental change*, 4(1), 37-48.
- Boko, M., Niang, I., Nyong, A., Vogel, A., Githeko, A., Medany, M., ... & Yanda, P. Z. (2018).
 Africa Climate Change 2007: Impacts, Adaptation and Vulnerability: Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change.Brisson, E., Demuzere, M., Willems, P., & van Lipzig, N. P. (2015).
 Assessment of natural climate variability using a weather generator. *Climate Dynamics*, 44(1-2), 495-508.
- Byass, P. (2009). Climate change and population health in Africa: where are the scientists?. *Global Health Action*, *2*(1), 2065.
- Cameron, A. C., & Trivedi, P. K. (2013). *Regression analysis of count data* (Vol. 53). Cambridge university press.

- Carlton, E. J., Eisenberg, J. N., Goldstick, J., Cevallos, W., Trostle, J., & Levy, K. (2013). Heavy rainfall events and diarrhea incidence: the role of social and environmental factors. *American journal of epidemiology*, *179*(3), 344-352.
- Carter, M. R., & Maluccio, J. A. (2003). Social capital and coping with economic shocks: an analysis of stunting of South African children. *World Development*, *31*(7), 1147-1163.
- Centers for Disease Control and Prevention. (2002). Heat-related deaths--four states, July-August 2001, and United States, 1979-1999. *MMWR: Morbidity and mortality weekly report*, 51(26), 567-570.
- Change, I. C. (2014). Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. *Cambridge University Press, Cambridge, UK and New York, NY*.
- Charles, D. (November 15, 2018). How Diseases Spread through the Fecal-Oral Route Proper Handwashing is you Best Defense [Blog post]. Retrieved from https://www.verywellhealth.com/what-is-the-fecal-oral-route-1760046.
- Chowdhury, F., Khan, I. A., Patel, S., Siddiq, A. U., Saha, N. C., Khan, A. I., ... & Ali, M. (2015). Diarrheal illness and healthcare seeking behavior among a population at high risk for diarrhea in Dhaka, Bangladesh. *PLoS One*, *10*(6), e0130105.
- Collier, P., Conway, G., & Venables, T. (2008). Climate change and Africa. Oxford Review of Economic Policy, 24(2), 337-353.
- Cong, R. G., & Brady, M. (2012). The interdependence between rainfall and temperature: copula analyses. *The Scientific World Journal*, 2012.

- Coxe, S., West, S. G., & Aiken, L. S. (2009). The analysis of count data: A gentle introduction to Poisson regression and its alternatives. *Journal of personality assessment*, 91(2), 121-136.
- Curtis, V., & Cairneross, S. (2003). Effect of washing hands with soap on diarrhoea risk in the community: a systematic review. *The Lancet infectious diseases*, *3*(5), 275-281.
- D'Acremont, V., Lengeler, C., Mshinda, H., Mtasiwa, D., Tanner, M., & Genton, B. (2009). Time to move from presumptive malaria treatment to laboratory-confirmed diagnosis and treatment in African children with fever. *PLoS medicine*, *6*(1), e252.
- Darrow, L. A., Hess, J., Rogers, C. A., Tolbert, P. E., Klein, M., & Sarnat, S. E. (2012). Ambient pollen concentrations and emergency department visits for asthma and wheeze. *Journal of Allergy and Clinical Immunology*, 130(3), 630-638.
- Datt, G., & Hoogeveen, H. (2003). El Niño or El Peso? Crisis, poverty and income distribution in the Philippines. *World Development*, *31*(7), 1103-1124.
- Dilaveris, P., Synetos, A., Giannopoulos, G., Gialafos, E., Pantazis, A., & Stefanadis, C. (2006). CLimate Impacts on Myocardial infarction deaths in the Athens TErritory: the CLIMATE study. *Heart*, 92(12), 1747-1751.
- Duflo, E., & Udry, C. (2004). Intrahousehold resource allocation in Cote d'Ivoire: Social norms, separate accounts and consumption choices (No. w10498).
 National Bureau of Economic Research.
- Egeru, A. (2012). Role of indigenous knowledge in climate change adaptation: A case study of the Teso Sub-Region, Eastern Uganda.

- Energy Sector GIS Working Group Uganda. (2017, August 11). Uganda Subregion Boundaries
 (2014) openAFRICA. Retrieved March 1, 2019, from https://africaopendata.org/dataset/subregion-boundaries-2014
- Food and Agriculture Organization (FAO) & World Food Programme (WFP) (2008, February 8). Assessment of the Impact of 2007 Floods on Food and Agriculture in Eastern and Northern Uganda. Retrieved March 1, 2019, from <u>http://www.fao.org/3/ah878e/ah878e00.htm</u>
- Franchini, M., & Mannucci, P. M. (2015). Impact on human health of climate changes. *European journal of internal medicine*, *26*(1), 1-5.

Funari, E., Manganelli, M., & Sinisi, L. (2012). Impact of climate change on waterborne diseases. *Annali dell'Istituto superiore di sanita*, 48, 473-487.

- Franks, P., Muennig, P., Lubetkin, E., & Jia, H. (2006). The burden of disease associated with being African-American in the United States and the contribution of socio-economic status. *Social science & medicine*, *62*(10), 2469-2478.
- Gardner, W., Mulvey, E. P., & Shaw, E. C. (1995). Regression analyses of counts and rates: Poisson, overdispersed Poisson, and negative binomial models. *Psychological bulletin*, *118*(3), 392.
- Gasper, R., Blohm, A., & Ruth, M. (2011). Social and economic impacts of climate change on the urban environment. Current Opinion in Environmental Sustainability, 3(3), 150-157.
- Gaurav, S. (2015). Are rainfed agricultural households insured? Evidence from five villages in Vidarbha, India. *World Development*, 66, 719-736.

- Goerre, S., Egli, C., Gerber, S., Defila, C., Minder, C., Richner, H., & Meier, B. (2007). Impact of weather and climate on the incidence of acute coronary syndromes. *International journal of cardiology*, 118(1), 36-40.
- Goodridge, D., Hawranik, P., Duncan, V., & Turner, H. (2012). Socioeconomic disparities in home health care service access and utilization: A scoping review. *International Journal of Nursing Studies*, 49(10), 1310-1319.
- Goosse, H., Barriat, P. Y., Loutre, M. F., & Zunz, V. (2010). *Introduction to climate dynamics and climate modeling*. Centre de recherche sur la Terre et le climat Georges Lemaître.
- Government Uganda. Ministry of Health. (2016). Annual Health Sector Performance Report. Kampala, Uganda. Retrieved from http://health.go.ug/content/annual-health-sectorperformance-report-201516
- Grazzini, G., Liumbruno, G. M., Pupella, S., Silvestri, A. R., Randi, V., Pascarelli, N., ... & Sambri, V. (2008). West Nile virus in Italy: a further threat to blood safety, a further challenge to the blood system. *Blood Transfus*, 6(4), 235-7.
- Greene, W. (2004). Fixed effects and bias due to the incidental parameters problem in the Tobit model. *Econometric reviews*, 23(2), 125-147.
- Griggs, D. J., & Noguer, M. (2002). Climate change 2001: the scientific basis. Contribution of working group I to the third assessment report of the intergovernmental panel on climate change. *Weather*, 57(8), 267-269.
- Guha-Sapir, D., Vos, F., Below, R., & Ponserre, S. (2012). Annual disaster statistical review 2011: the numbers and trends. Centre for Research on the Epidemiology of Disasters (CRED).

- Haines, A., Kovats, R. S., Campbell-Lendrum, D., & Corvalán, C. (2006). Climate change and human health: impacts, vulnerability and public health. *Public health*, 120(7), 585-596.
- Hales, S., De Wet, N., Maindonald, J., & Woodward, A. (2002). Potential effect of population and climate changes on global distribution of dengue fever: an empirical model. *The Lancet*, 360(9336), 830-834.
- Hales, S., Weinstein, P., & Woodward, A. (1996). Dengue fever epidemics in the South Pacific: driven by El Niño southern oscillation?. *The Lancet*, 348(9042), 1664-1665.
- Hanjra, M. A., & Qureshi, M. E. (2010). Global water crisis and future food security in an era of climate change. *Food Policy*, 35(5), 365-377.
- Hansen, J., Sato, M., Ruedy, R., Lo, K., Lea, D. W., & Medina-Elizade, M. (2006). Global temperature change. *Proceedings of the National Academy of Sciences*, 103(39), 14288-14293.
- Hashizume, M., Armstrong, B., Hajat, S., Wagatsuma, Y., Faruque, A. S., Hayashi, T., & Sack, D.
 A. (2008). The effect of rainfall on the incidence of cholera in Bangladesh. *Epidemiology*, 19(1), 103-110.
- Hennessy, K. B., Fitzharris, B., Bates, B. C., Harvey, N., Howden, M., Hughes, L., ... & Warrick,R. (2007). Australia and New Zealand: climate change 2007: impacts, adaptation and vulnerability: contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change.
- Hoddinott, J., & Kinsey, B. (2001). Child growth in the time of drought. Oxford Bulletin of Economics and statistics, 63(4), 409-436.
- Hutton, G. (2011). The economics of health and climate change: key evidence for decision making. *Globalization and health*, 7(1), 18.
- IPCC. (2012). Managing the risks of extreme events and disasters to advance climate change adaptation: special report of the intergovernmental panel on climate change. Cambridge University Press.
- Ingrid, K. (November 16, 2018). Infectious Diseases Causes, Include Viruses, Bacteria and Parasites [Blog post]. Retrieved from https://www.verywellhealth.com/diarrheal-diseases-101-1958810.
- International Federation of Red Cross and Red Crescent Societies. (2010, July 31). Uganda: Floods & Landslides in Eastern Uganda. Retrieved March 1, 2019, from http://www.ifrc.org/docs/appeals/10/MDRUG015fr.pdf.
- Irish Aid. (2017). UGANDA CLIMATE ACTION REPORT FOR 2016. [online] Available at: https://www.irishaid.ie/media/irishaid/allwebsitemedia/30whatwedo/climatechange/Ugand a-Country-Climate-Action-Reports-

2016.pdf?fbclid=IwAR0Tc8ll3JZ0aEyxWomS3SyCsjWNoRObFoX2KzO3BbkGmQAX MjYv6w-5tM [Accessed 17 Feb. 2019].

- Ito, T., & Kurosaki, T. (2009). Weather risk, wages in kind, and the off-farm labor supply of agricultural households in a developing country. American journal of agricultural economics, 91(3), 697-710.
- Jack, W., & Suri, T. (2014). Risk sharing and transactions costs: Evidence from Kenya's mobile money revolution. *American Economic Review*, 104(1), 183-223.

Jacob, D. J., & Winner, D. A. (2009). Effect of climate change on air quality. *Atmospheric environment*, 43(1), 51-63.

- Judkins, R. R., Fulkerson, W., & Sanghvi, M. K. (1993). The dilemma of fossil fuel use and global climate change. *Energy & fuels*, 7(1), 14-22
- Karl, T. R., Melillo, J. M., Peterson, T. C., & Hassol, S. J. (2009). Global Climate Change Impacts in the United States: Highlights.
- Kazianga, H., & Udry, C. (2006). Consumption smoothing? Livestock, insurance and drought in rural Burkina Faso. *Journal of Development economics*, 79(2), 413-446.
- Khasnis, A. A., & Nettleman, M. D. (2005). Global warming and infectious disease. Archives of medical research, 36(6), 689-696.
- Kim, Y., & Yang, B. (2011). Relationship between catastrophic health expenditures and household incomes and expenditure patterns in South Korea. *Health policy*, 100(2-3), 239-246.
- Kochar, A. (1999). Smoothing consumption by smoothing income: hours-of-work responses to idiosyncratic agricultural shocks in rural India. *Review of Economics and Statistics*, 81(1), 50-61.
- Kovats, S., Wolf, T., & Menne, B. (2004). Heatwave of August 2003 in Europe: provisional estimates of the impact on mortality. *Eurosurveillance Weekly*, 8(11), 11.
- Kreft, S., Eckstein, D., & Melchior, I. (2017). Global Climate Risk Index: Who Suffers Most From Extreme Weather Events? Weather-related Loss Events in 2015 and 1996 to 2015. *Germanwatch Nord-Süd Initiative eV.* Retrieved on 17th February 2020, from https://reliefweb.int/report/world/global-climate-risk-index-2017-who-suffers-mostextreme-weather-events-weather-related

- Kurosaki, T. (2006). Consumption vulnerability to risk in rural Pakistan. *The journal of development studies*, 42(1), 70-89.
- Landrigan, P. J., Fuller, R., Acosta, N. J., Adeyi, O., Arnold, R., Baldé, A. B., ... & Chiles, T. (2018). The Lancet Commission on pollution and health. *The Lancet*, *391*(10119), 462-512.
- Landrigan, P. J. (2004). Children as a vulnerable population. *International journal of occupational medicine and environmental health*, *17*(1), 175-178.
- Labbé, J., Ford, J. D., Berrang-Ford, L., Donnelly, B., Lwasa, S., Namanya, D. B., ... & IHACC
 Research Team. (2016). Vulnerability to the health effects of climate variability in rural southwestern Uganda. *Mitigation and adaptation strategies for global change*, 21(6), 931-953.
- Lafferty, K. D. (2009). The ecology of climate change and infectious diseases. *Ecology*, *90*(4), 888-900.
- Lim, Y. H., Hong, Y. C., & Kim, H. (2012). Effects of diurnal temperature range on cardiovascular and respiratory hospital admissions in Korea. *Science of The Total Environment*, 417, 55-60.
- Lin, S., Luo, M., Walker, R. J., Liu, X., Hwang, S. A., & Chinery, R. (2009). Extreme high temperatures and hospital admissions for respiratory and cardiovascular diseases. *Epidemiology*, 20(5), 738-746.
- Lohmann, S., & Lechtenfeld, T. (2015). The effect of drought on health outcomes and health expenditures in rural Vietnam. *World development*, 72, 432-448.

- Louis, M. E. S., & Hess, J. J. (2008). Climate change: impacts on and implications for global health. *American journal of preventive medicine*, *35*(5), 527-538.
- Lord, D., Park, B. J., & Model, P. G. (2012). Negative Binomial Regression Models and Estimation Methods. *Probability density and likelihood functions*, 1-15.
- Lowder, S. K., Bertini, R., Karfakis, P., & Croppenstedt, A. (2016). *Transformation in the size and distribution of farmland operated by household and other farms in select countries of sub-Saharan Africa* (No. 310-2016-5344).
- Maccini, S., & Yang, D. (2009). Under the weather: Health, schooling, and economic consequences of early-life rainfall. *American Economic Review*, *99*(3), 1006-26.
- Mawa, R., & Lawoko, S. (2018). Malnutrition Among Children Under Five Years in Uganda. *American Journal of Health Research*, 6(2), 56-66.
- McCalla, A. F. (2009). World food prices: Causes and consequences. *Canadian Journal of Agricultural Economics/Revue canadienne d'agroeconomie*, 57(1), 23-34.
- McCarthy, J. J. (2001). Climate change 2001: impacts, adaptation, and vulnerability: contribution of Working Group II to the third assessment report of the Intergovernmental Panel on Climate Change. Cambridge University Press. doi.org/10.1002/joc.775>.
- McIver, L., Hashizume, M., Kim, H., Honda, Y., Pretrick, M., Iddings, S., & Pavlin, B. (2014). Assessment of climate-sensitive infectious diseases in the Federated States of Micronesia. *Tropical medicine and health*.
- McMichael, A. J. (2003). *Climate change and human health: risks and responses*. World Health Organization.

- McMichael, A. J. (2013). Globalization, climate change, and human health. *New England Journal of Medicine*, *368*(14), 1335-1343.
- McMichael, A. J. (2013). Globalization, climate change, and human health. *New England Journal of Medicine*, *368*(14), 1335-1343.
- Mendelsohn, R., Dinar, A., & Williams, L. (2006). The distributional impact of climate change on rich and poor countries. *Environment and Development Economics*, 11(2), 159-178.
- Medina-Ramón, M., Zanobetti, A., Cavanagh, D. P., & Schwartz, J. (2006). Extreme temperatures and mortality: assessing effect modification by personal characteristics and specific cause of death in a multi-city case-only analysis. *Environmental health perspectives*, 1331-1336.
- Meehl, G. A., & Tebaldi, C. (2004). More intense, more frequent, and longer lasting heat waves in the 21st century. *Science*, 305(5686), 994-997.
- Mills, B., & Andrey, J. (2002). Climate change and transportation: potential interactions and impacts. The Potential Impacts of Climate Change on Transportation, [US] DOT Center for Climate Change and Environmental Forecasting (http://climate. volpe. dot. gov/workshop1002/mills. doc), 77-88.
- Mubiru, D. N., Komutunga, E., Agona, A., Apok, A., & Ngara, T. (2012). Characterising agrometeorological climate risks and uncertainties: Crop production in Uganda. South African Journal of Science, 108(3-4), 108-118.
- Mukamal, K. J., Wellenius, G. A., Suh, H. H., & Mittleman, M. A. (2009). Weather and air pollution as triggers of severe headaches. *Neurology*, *72*(10), 922-927.

- Musengimana, G., Mukinda, F., Machekano, R., & Mahomed, H. (2016). Temperature variability and occurrence of diarrhoea in children under five-years-old in Cape Town metropolitan sub-districts. International journal of environmental research and public health, 13(9), 859.
- Muttarak, R., Lutz, W., Jiang, L. (2016). What can demographers contribute to the study of vulnerability? *Vienna Yearbook of Population Research* 2015(13) 1-13.
- Nimusiima, A., Basalirwa, C. P. K., Majaliwa, J. G. M., Otim-Nape, W., Okello-Onen, J., Rubaire-Akiiki, C., ... & Ogwal-Byenek, S. (2013). Nature and dynamics of climate variability in the Uganda cattle corridor. African Journal of Environmental Science and Technology, 7(8), 770-782.
- Ogundari, K., & Abdulai, A. (2014). Determinants of household's education and healthcare spending in Nigeria: Evidence from survey data. African Development Review, 26(1), 1-14.
- Okunade, A. A., Suraratdecha, C., & Benson, D. A. (2010). Determinants of Thailand household healthcare expenditure: the relevance of permanent resources and other correlates. Health Economics, 19(3), 365-376.
- Onah, M. N., & Govender, V. (2014). Out-of-pocket payments, health care access and utilisation in south-eastern Nigeria: a gender perspective. PLoS One, 9(4), e93887.
- Osgood, D. W. (2000). Poisson-based regression analysis of aggregate crime rates. *Journal of quantitative criminology*, *16*(1), 21-43.

- Parry, M. L., Rosenzweig, C., Iglesias, A., Livermore, M., & Fischer, G. (2004). Effects of climate change on global food production under SRES emissions and socio-economic scenarios. *Global Environmental Change*, 14(1), 53-67.
- Paternoster, R., & Brame, R. (1997). Multiple routes to delinquency? A test of developmental and general theories of crime. *Criminology*, 35(1), 49-84.
- Patt, A. G., Tadross, M., Nussbaumer, P., Asante, K., Metzger, M., Rafael, J., ... & Brundrit, G. (2010). Estimating least-developed countries' vulnerability to climate-related extreme events over the next 50 years. *Proceedings of the National Academy of Sciences*, 107(4), 1333-1337.
- Pascual, M., Bouma, M. J., & Dobson, A. P. (2002). Cholera and climate: revisiting the quantitative evidence. *Microbes and Infection*, 4(2), 237-245.
- Petit, J. R., Jouzel, J., Raynaud, D., Barkov, N. I., Barnola, J. M., Basile, I., ... & Delmotte, M. (1999). Climate and atmospheric history of the past 420,000 years from the Vostok ice core, Antarctica. *Nature*, 399(6735), 429-436.
- Prasad, R., Garg, R., & Hosmane, G. B. (2012). Biomass fuel exposure and respiratory diseases in India. *Bioscience trends*, 6(5), 219-228.
- Prina, S. (2015). Banking the poor via savings accounts: Evidence from a field experiment. *Journal of Development Economics*, *115*, 16-31.
- Prince, P. B., Rapoport, A. M., Sheftell, F. D., Tepper, S. J., & Bigal, M. E. (2004). The effect of weather on headache. *Headache: The Journal of Head and Face Pain*, 44(6), 596-602.

- Rahman, M. I. U. (2013). Climate change: A theoretical review. Interdisciplinary Description of Complex Systems: INDECS, 11(1), 1-13.
- Randall, D. A., Wood, R. A., Bony, S., Colman, R., Fichefet, T., Fyfe, J., ... & Stouffer, R. J. (2007).
 Climate models and their evaluation. In *Climate change 2007: The physical science basis. Contribution of Working Group I to the Fourth Assessment Report of the IPCC (FAR)* (pp. 589-662). Cambridge University Press.
- Reliefweb. (2010, March). East Africa: Floods. Retrieved March 1, 2019, from https://reliefweb.int/disaster/ls-2010-000040-uga.
- Rose, E. (1999). Consumption smoothing and excess female mortality in rural India. *Review of Economics and statistics*, 81(1), 41-49.
- Satterthwaite, F. E. (1946). An approximate distribution of estimates of variance components. *Biometrics bulletin*, 2(6), 110-114.
- Sellman, J., & Hamilton, J. D. (2007). Global climate change and human health. *Minnesota medicine*, 90(3), 47-50.
- Simpson, R., Williams, G., Petroeschevsky, A., Best, T., Morgan, G., Denison, L., ... & Neller, A. (2005). The short-term effects of air pollution on daily mortality in four Australian cities. *Australian and New Zealand journal of public health*, 29(3), 205-212.
- Smith, J. (2010). Understanding climate variability and climate change. *Weather and Climate*, *1*, 1-25.

- Solomon, S., Qin, D., Manning, M., Chen, Z., Marquis, M., Averyt, K. B., ... & Miller, H. L. (2007). Contribution of working group I to the fourth assessment report of the intergovernmental panel on climate change, 2007.
- Stanke, C., Kerac, M., Prudhomme, C., Medlock, J., & Murray, V. (2013). Health effects of drought: a systematic review of the evidence. *PLoS currents*, 5.
- Stephenson, D. B., Diaz, H. F., & Murnane, R. J. (2008). Definition, diagnosis, and origin of extreme weather and climate events (Vol. 348). Cambridge University Press: New York.
- Szyszkowicz, M., Kaplan, G., Grafstein, E., & Rowe, B. (2009). Emergency department visits for migraine and headache: a multi-city study. *International journal of occupational medicine* and environmental health, 22(3), 235-242.
- Tesfay, G. B., & Abidoye, B. (2019). Shocks in food availability and intra-household resources allocation: evidence on children nutrition outcomes in Ethiopia. Agricultural and Food Economics, 7(1), 3.
- Thomas, D. S., Twyman, C., Osbahr, H., & Hewitson, B. (2007). Adaptation to climate change and variability: farmer responses to intra-seasonal precipitation trends in South Africa. *Climatic change*, *83*(3), 301-322.
- Thornton, P. K., Jones, P. G., Alagarswamy, G., & Andresen, J. (2009). Spatial variation of crop yield response to climate change in East Africa. *Global Environmental Change*, 19(1), 54-65.
- Tosam, M. J., & Mbih, R. A. (2015). Climate change, health, and sustainable development in Africa. *Environment, development and sustainability*, *17*(4), 787-800. doi:10.1007/s10668-014-9575-0.

Townsend, R. M. (1995). Consumption insurance: An evaluation of risk-bearing systems in lowincome economies. *Journal of Economic Perspectives*, 9(3), 83-102.

Uganda Bureau of Statistics. (2014). National population and housing census 2014.

- Uganda Investment Authority. (2016). Teso Investment Profile 2016. Republic of Uganda, Kampala. Retrieved from https://www.ugandainvest.go.ug/wpcontent/uploads/2016/02/Teso-Investment-Profile.pdf
- Uganda Ministry of Water and Environment. (2015). Economic Assessment of the Impacts of Climate Change in Uganda. Kampala, Uganda. Anil Markandya, Courtenay Cabot-Venton & Olivier Beucher.
- United Nations Development Programme. (2015). Uganda Human Development Report: Unlocking the Development Potential of Northern Uganda. Kampala, Uganda. Retrieved from

http://www.ug.undp.org/content/dam/uganda/docs/UNDPUg2015_UgandaHDR2015.pdf.

- United Nations Framework Convention on Climate Change: Fact sheet: Climate Change science the status of Climate Change science today. UNFCC, 2011. Retrieved from http://unfccc.int/files/press/backgrounders/application/pdf/press_factsh_science.pdf.
- Van den Brink, M., Bandell-Hoekstra, E. N. G., & Abu-Saad, H. H. (2001). The occurrence of recall bias in pediatric headache: a comparison of questionnaire and diary data. *Headache: The Journal of Head and Face Pain*, 41(1), 11-20.

Verner, G., Schütte, S., Knop, J., Sankoh, O., & Sauerborn, R. (2016). Health in climate change research from 1990 to 2014: positive trend, but still underperforming. *Global health action*, *9*(1), 30723.

- Vu, L., & Glewwe, P. (2011). Impacts of rising food prices on poverty and welfare in Vietnam. Journal of Agricultural and Resource Economics, 14-27.
- Wagstaff, A., & Doorslaer, E. V. (2003). Catastrophe and impoverishment in paying for health care: with applications to Vietnam 1993–1998. *Health economics*, 12(11), 921-933.
- Wheeler, T., & Von Braun, J. (2013). Climate change impacts on global food security. *Science*, 341(6145), 508-513.
- World Bank Group, Climate Change Knowledge Portal (2019). Accessed January 19, 2019 at http://sdwebx.worldbank.org/climateportal/
- World Bank. (2018, May). Uganda Economic Update, 11th Edition. FINANCING GROWTH
 AND DEVELOPMENT: Options for raising more domestic revenues. Retrieved February
 28, 2019, from
 http://documents.worldbank.org/curated/en/425631526323380885/pdf/126184-WPPUBLICFinalReportUgandaEconomicUpdate.pdf?fbclid=IwAR1j_v4rHH7VdVAgn3N2CU

F-1sH0U5e_iCT6tWzFKpM4MX74HgYC7UqcnsQ

- World Bank. (2012b). The welfare effects of extreme weather events Insights from three APEC case studies. Washington, DC.
- World Bank. (2016). The Uganda poverty assessment report 2016. Farms, cities and good fortune: assessing poverty reduction in Uganda from 2006 to 2013. Kampala, Uganda. Retrieved

from http://pubdocs.worldbank.org/en/381951474255092375/pdf/Uganda-Poverty-Assessment-Report-2016.pdf

- World Bank. (2016). Uganda Economic Update. From smart budgets to smart returns: unleashing the power of public investment management. Kampala, Uganda. Retrieved from http://documents.worldbank.org/curated/en/859951467989540438/pdf/106178-REPLACEMENT-Uganda-Economic-Update.pdf
- World Development Indicators, Washington, DC. The World Bank Group. Accessed February 6, 2019 at http://datatopics.worldbank.org/world-development-indicators/.
- World Health Organization. (2018). Country Cooperation Strategy Uganda. Retrieved from https://apps.who.int/iris/bitstream/handle/10665/136975/ccsbrief_uga_en.pdf;jsessionid=D 0FB4B3F91CF77ED4A468AF472C94759?sequence=1
- World Health Organization. (2006). Constitution of the World Health Organization. *Basic Documents. 45th edition Supplement*.
- World Health Organization. (2009). Diarrhea: Why children are still dying and what can be done. Retrieved March 17, 2019, from https://apps.who.int/iris/bitstream/handle/10665/44174/9789241598415_eng.pdf;jsessioni d=F5E423F3D84AD79B930963F33EB31408?sequence=1
- World Health Organization. (2003). How much disease would climate change cause? Retrieved from https://www.who.int/globalchange/environment/en/chapter7.pdf.
- WHO (World Health Organization. (2018). Uganda's health sector grows despite challenges. Retrieved from https://afro.who.int/news/ugandas-health-sector-grows-despite-challenges.

- Yang, A. C., Fuh, J. L., Huang, N. E., Shia, B. C., Peng, C. K., & Wang, S. J. (2011). Temporal associations between weather and headache: analysis by empirical mode decomposition. *PloS one*, 6(1), e14612.
- Zhou, G., Minakawa, N., Githeko, A. K., & Yan, G. (2004). Association between climate variability and malaria epidemics in the East African highlands. Proceedings of the National Academy of Sciences of the United States of America, 101(8), 2375-2380.

Appendix

Photos from the field – Teso sub-region

Water vendors



Appendix 1: Vendor going to fetch water for sale



Appendix 2: Vendor carrying jerry cans of water for sale

Collecting outpatient data archives



Appendix 3: Checking through outpatient records



Appendix 4: At Amuria health data office, the lady (biostatistician) helps to retrieve data files

	OUTPAT	MORIN A PRELL 2010 1. OPD ATTENDANCE AND LABORATORY TESTS TOTALS FOR THE MONTH											
	Category	0-4 years	1.4	and over		LABORATORY TESTS	No. of tests	done	No. Pos	tive			
New attendance			Malo I	Female A	Male F	emale	Category	0.4 yrs	S23 S	363 -	21		
	Re-attendance Total Attendance		00	01	00 4	01	TB sputum	00 0	29 1		5 120		
	Referrals to unit		29	35 1	44 4	82	Pap smear	NTA	VIA	NA	SA		
	2 01	ITRATIENT D	00	03	02	01	Other Lab Tests - HV	00 10					
	Diagnosis	JIPATIENT D	0-4 ypa	DSES	5 and o	-		0-4 yrs Male Female		5 and ou	or T		
	Enidemic Prone Die		Male	Female	Male	Female	Diagnosis Inchie Disease			Male Female			
	01 Acute flaccid pa	ralysis	De	De	De	00	Other Infectious/Communicable Disease	02	00	50	88		
	02 Cholera		00	00	00	00	14 Diarrhea-Acute	DH	03	DI	00		
	03 Dysentery		00	01	01	20	15 Diamea- Persistent	00	00	00	00		
	04 Guinea worm		00	00	00	00	16 ENT conditions	02	00	DU	00		
	06 Measles	ingueoceai)	00	00	00	00	17 Eye conditions 18 Severally Transmitted Infection (STI)	00	00	00	00		
	07 Tetanus (neonat	al)(0 -28 days apro)	00	00	00	00	19 Unnary Tract Infections (UTI)	11	05	21	91		
	08 Plague		00	00	00	00	20 Intestinal Worms	00	00	00	00		
	09 Rabies		00	00	00	00	21 Leprosy	00	00	00	00		
	10 Yellow Fever		00	00	00	00	22 Malaria	190	177	142	268		
	11Other Viral Haem	orthagic Fevers	00	00	00	00	23 Other types of meningitis	00	00	00	18		
	infectious disease		00	00	00	00	25 Preumonia	011	OL	00	00		
	(Specify)		-	-			26 Schistosomiasis	00	00	00	02		
	Maternal and Perinat	al Diseases		-	-		27 Onchocerciasis	00	00	00	00		
	35 Abortions		00	00	00	DS	28 Skin Diseases 29 Tuberculosis (New cases)	09	00	04	01		
	35 Matana in program.y	- manager	00	00	00	01	30 Typhoid Fever	00	00	03	05		
	38 Obstructed labour		00	00	00	00	31 Tetanus (over 28 days age)	00	00	00	00		
	P Haemorrhage related	to pregnancy (APH	00	00	00	01	32 Sleeping sickness	00	00	00	OD		
	40 Perioatal conditions (in	n new borns 0 - 28	0.	00	00	00	33 Pelvic Inflammatory Disease (PID)	00	00	00	13		
	days)			07	00	00	34 Deaths in OPD	00	00	00	00		
	Non-communicable d	liseases	-			1 1 1 1	More Non-communicable diseases	1	1	1	1		
	41 Anaemia		33	43	12	12	54 Other forms of mental illness	04	00	02	02		
	42 Asthma	nditions	00	00	00	00	55 Other cardiovascular diseases	00	01	02	00		
	44 Diabetes mellitus		00	00	07	0.5	56 Severe Malnutrition (Marasmus, Kwashiorkor and Marasmic-kwash)	01	02	00	00		
45 Gastro Intestinal disorders (non-Infective)			08	10	16	79	57 Low weight for age	00	00	00	00		
	46 Hyportansion		00	00	04	18	59 Injuries= (Trauma due to other causes)-	00	02	12	00		
	47 Anxiety desorders	00	00	00	00	60 Animal/ snakes bites	00	01	12	04			
48 Manue			00	00	00	00	61 Other diagnoses (priority diseases for	00	0.	000	00		
	and to subject to a second		00	00	00	00	District)	00	00	00	00		
			00	00	00	DD	All others	0.0	1.	21	70		
	51 Alcohol and Drug abus	referen	00	00	00	00	Total Diagnoses	314	30	1 31-	1794		
52 Childhood Mental Disorders			00	00	00	00	La mais de la casa de la cas	010	150	100			

Appendix 5: Copy of Health Information Management System (HMIS) form



Appendix 6: Map of Uganda showing the distribution of Uganda National Panel Survey (UNPS) households *Source: Adopted from UNPS Wave 3 Report (2011)*

Section 5: Health

Ask the following	questions	about all	Imembers	of the	household	(usual and	d regular)
Ask the following	questions	aboutai	i illettibeta		nousenoiu	(usual an	. regular).

	INTERVIEWER:	1	During the	For how	For how many	Can yo	u	Was anyone	Why was no	Where did [NAME] go for	Distance	What was the	CODES FOR COL7	
P E R S O N I D	IS (NAME) ANSWERING FOR HMSELF OR HERSELF? 1= Yes (>>4) 2= No	WHAT IS THE ID CODE OF THE PERSON RESPONDING FOR [NAME]?	past 30 days, did [NAME] suffer from any illness or injury? 1= Yes 2= No (>> NEXT PERSON)	many days did [[NAME]] suffer due to illness or injury during the past 30 days?	days did [NAME] have to stop doing [NAME]'susual activities due to linesso rinjuny during the past 30 days? DAYS SHOULD BE LESS THAN OR EQUAL TO COL 5.	descrit symptot that [N primari suffere tothe m injury of the pasa days? <i>RECO T(C</i> <i>SYMM</i> <i>CO</i>) <i>SEE C</i> <i>SYM</i>	ve the vmms AME] ly d due anajor or during st 30 RD UP D 2 PTOM DES CODES VIGHT	consulted (e.g. a doctor, nurse, pharmacist or traditional healer) for the major illness/injury during the past 30 days? 1= Yes (>> 10) 2= No	one consulted for the major illness? SEE CODES AT RIGHT >NEXT PERSON]	the first consultation during the past 30 days? PUBLIC SECTOR 1= Government hospilal 2= Government hospilal 2= Government Community Based Distributor PRIVATE SECTOR 5= Private hospital 6= Pharmacyi dayshop 7= Private Doctor/Nurse/Midwife/Clinic 8= Outerach 9= NGO Community Based Distributor OTHER SOURCE 10= Shop 11= Religious Institution 12= Friend/ Relative 13= Traditional Healer 96= Other (specify)	to the place where this treatment was sought for in km?	cost of this consultation, including any medicine prescribed even if purchased elsewhere?	I = Utamhosa (acute) 2 = Dianhosa (acute) 2 = Dianhosa (chronio, 1 month or more) 3 = Weight loss (major) 4 = Fever (acute) 6 = Fever (acute) 6 = Wound 7 = Skin rash 8 = Weakness 9 = Severe headache 10 = Fairding 11 = Chills (feeling hot and odl) 12 = Voonting 13 = Cough 13 = Cough 14 = Troductive cough 15 = Cough 16 = Cough 16 = Cough 16 = Cough 16 = Cough 16 = Cough 17 = Cough 16 = Cough 17 = Cough 18 = Cough 19 = Co	
				DAYS	DAYS	74	70			40	KMS	SHILLINGS	are too costly 6= No qualified staff	
1	2	3	4	5	6	/A	/B	8	9	10	11	12	present 7= Staff attitude not	
01													good 8= Too busy / long	
02													waiting time 9= Facility is	
03													inaccessible 10= Facility is closed	
05													11= Facility is destroyed	
06													12= Drugs not available 96= Other (specify)	
07														
08														
09														
10														

Appendix 7: UNPS Questionnaire - Health module

Source: Adopted from UNPS (2011) Household Questionnaire